### CLINICAL GUIDELINE



# Treatment of Anemia in Patients With Heart Disease: A Clinical Practice Guideline From the American College of Physicians

Amir Qaseem, MD, PhD, MHA; Linda L. Humphrey, MD, MPH; Nick Fitterman, MD; Melissa Starkey, PhD; and Paul Shekelle, MD, PhD, for the Clinical Guidelines Committee of the American College of Physicians\*

**Description:** The American College of Physicians (ACP) developed this guideline to present the evidence and provide clinical recommendations on the treatment of anemia and iron deficiency in adult patients with heart disease.

**Methods:** This guideline is based on published literature in the English language on anemia and iron deficiency from 1947 to July 2012 that was identified using MEDLINE and the Cochrane Library. Literature was reassessed in April 2013, and additional studies were included. Outcomes evaluated for this guideline included mortality; hospitalization; exercise tolerance; quality of life; and cardiovascular events (defined as myocardial infarction, congestive heart failure exacerbation, arrhythmia, or cardiac death) and harms, including hypertension, venous thromboembolic events, and ischemic cerebrovascular events. The target audience for this guideline includes all clinicians, and the target patient population is anemic or iron-

deficient adult patients with heart disease. This guideline grades the evidence and recommendations using the ACP's clinical practice guidelines grading system.

**Recommendation 1:** ACP recommends using a restrictive red blood cell transfusion strategy (trigger hemoglobin threshold of 7 to 8 g/dL compared with higher hemoglobin levels) in hospitalized patients with coronary heart disease. (Grade: weak recommendation; low-quality evidence)

**Recommendation 2:** ACP recommends against the use of erythropoiesis-stimulating agents in patients with mild to moderate anemia and congestive heart failure or coronary heart disease. (Grade: strong recommendation; moderate-quality evidence)

Ann Intern Med. 2013;159:770-779. For author affiliations, see end of text. www.annals.org

A nemia is common in patients with heart disease. It is present in approximately one third of patients with congestive heart failure (CHF) and 10% to 20% of patients with coronary heart disease (CHD) (1–3). The cause of anemia in heart disease is not fully understood. Several factors probably contribute, including iron deficiency, comorbid chronic kidney disease, blunted erythropoietin production, hemodilution, aspirin-induced gastrointestinal blood loss, use of renin–angiotensin–aldosterone system blockers, cytokine-mediated inflammation (anemia of chronic disease), and gut malabsorption with consequent nutritional deficiency.

Anemia can worsen cardiac function and is associated with poor outcomes, including increased risk for hospitalization and death, decreased exercise capacity, and poor

#### See also:

Related article
Summary for Patients I-32
Web-Only
CME quiz

quality of life. However, it is not clear whether anemia directly and independently leads to these poor outcomes or whether it reflects more severe underlying illness (4-6). Treatments for anemia in patients with heart disease include erythropoiesis-stimulating agents (ESAs), red blood cell (RBC) transfusion, and iron replacement, although it is unclear whether these strategies improve outcomes.

The purpose of this American College of Physicians (ACP) guideline is to present current evidence on the treatment of anemia in patients with heart disease or iron deficiency. The target audience for this guideline includes all clinicians, and the target patient population is anemic or iron-deficient adult patients with heart disease. These recommendations are based on a background paper (7) and a systematic evidence review sponsored by the U.S. Department of Veterans Affairs (8).

#### **M**ETHODS

This guideline is based on a systematic evidence review and summary paper (7, 8) that addressed the following key questions related to the treatment of anemia in patients with heart disease:

1. In patients with CHF or CHD, what are the health benefits and harms of treating anemia with RBC transfusions?

\* This paper, written by Amir Qaseem, MD, PhD, MHA; Linda L. Humphrey, MD, MPH; Nick Fitterman, MD; Melissa Starkey, PhD; and Paul Shekelle, MD, PhD, was developed for the Clinical Guidelines Committee of the American College of Physicians. Individuals who served on the Clinical Guidelines Committee from initiation of the project until its approval were: Paul Shekelle, MD, PhD (*Chair*); Roger Chou, MD; Molly Cooke, MD; Paul Dallas, MD; Thomas D. Denberg, MD, PhD; Paul Dallas, MD; Nick Fitterman, MD; Mary Ann Forciea, MD; Linda L. Humphrey, MD, MPH; Tanveer P. Mir, MD; Holger J. Schünemann, MD, PhD; Donna E. Sweet, MD; and Timothy Wilt, MD, MPH. Approved by the ACP Board of Regents on 30 July 2013.

**770** © 2013 American College of Physicians

Downloaded From: http://annals.org/ on 10/14/2014 I his article has been corrected. I he specific correction appears on the last page of this document. The original version (PDF) is available at www.annals.org.

## Table 1. The American College of Physicians Guideline Grading System\*

Quality of Evidence	Strength of Recommendation		
Evidence	Benefits Clearly Outweigh Risks and Burden or Risks and Burden Clearly Outweigh Benefits	Benefits Finely Balanced With Risks and Burden	
High	Strong	Weak	
Moderate	Strong	Weak	
Low	Strong	Weak	
	Insufficient evidence to determin	e net benefits or risks	

\* Adopted from the classification developed by the GRADE (Grading of Recommendations Assessment, Development, and Evaluation) workgroup.

2. In patients with CHF or CHD, what are the health benefits and harms of treating anemia with ESAs?

3. In patients with CHF or CHD, what are the health benefits and harms of using iron to treat iron deficiency with or without anemia?

The systematic evidence review was conducted by the Evidence-based Synthesis Program Center at the Portland Veterans Affairs Medical Center. The literature search included English-language studies published from 1947 to July 2012 identified by using MEDLINE and the Cochrane Library. Additional information from the search came from systematic reviews; reference lists of pertinent studies, reviews, and editorials; by consulting experts and ClinicalTrials.gov; and by contacting pharmaceutical companies directly. In April 2013, the literature was reassessed, and 2 studies originally reported as in-progress were subsequently published and are included in this review (9, 10). Two reviewers assessed study quality according to a Cochrane protocol, and the Cochran Q test and  $I^2$  statistic were used to assess statistical heterogeneity (11). The outcomes of interest included mortality (all-cause and diseasespecific), hospitalization (all-cause and disease-specific), exercise tolerance (any metric, most commonly the New York Heart Association [NYHA] functional class and the 6-minute walk test), quality of life, and cardiovascular events (myocardial infarction [MI], exacerbation of heart failure, and need for revascularization). More details of the methods can be found in the article and full report (7, 8).

This guideline rates the evidence and recommendations by using ACP's guideline grading system (Table 1). Details of the ACP guideline development process can be found in the methods paper (12).

#### BENEFITS OF TREATMENT OF ANEMIA WITH RBC TRANSFUSIONS

# Medical and Surgical Patients Combined *Mortality*

Low-quality evidence from 6 studies of medical and noncardiac surgical patients (10, 13–17) assessed the effect

www.annals.org

of RBC transfusions to treat anemia in patients with heart disease and showed no mortality benefit for liberal RBC transfusion (hemoglobin level >10 g/dL) compared with restrictive RBC transfusion (hemoglobin level <10 g/dL) (relative risk [RR], 0.94 [95% CI, 0.61 to 1.42];  $I^2 = 16.8\%$ ). A recent randomized pilot trial of patients with unstable and stable CHD having cardiac catheterization found that a liberal transfusion strategy was associated with fewer major cardiac events and deaths. The group assigned to the restrictive transfusion strategy was older and had more patients with unstable CHD, but inclusion of these results in the meta-analysis still did not result in a statistically significant improvement in outcomes (10).

#### Cardiovascular Events

Low-quality evidence (13–17) showed that liberal RBC transfusions were associated with reduced cardiovascular events (RR, 0.64 [CI, 0.38 to 1.09];  $I^2 = 0.0\%$ ), although the data were not statistically significant.

#### Exercise Tolerance and Duration

Evidence was insufficient to determine the effect of RBC transfusions on exercise tolerance and duration.

#### Quality of Life

Evidence was insufficient to determine the effect of RBC transfusions on quality of life.

#### Nonsurgical Patients

#### Mortality

Low-quality evidence from 3 trials (16, 18, 19) showed no mortality benefit with a higher RBC transfusion threshold in nonsurgical patients with acute MI or known ischemic heart disease. One study of 838 euvolemic, nonbleeding, critically ill patients with hemoglobin levels less than 9 g/dL defined restrictive transfusion as a hemoglobin threshold of 7 g/dL with goal hemoglobin levels of 7 to 9 g/dL and a liberal strategy threshold as 10 g/dL with a goal of 10 to 12 g/dL (19). The study did not find any statistically significant difference between the 2 groups in hospital mortality or at 30 days after treatment. Post hoc subgroup analysis of patients with ischemic heart disease showed a nonstatistically significant higher mortality in the restrictive transfusion group (30-day mortality of 21.2% vs. 26.1% for liberal vs. restrictive strategies, respectively; P = 0.38) (16). The other study (18) of 45 patients with acute MI and hematocrit less than 30 defined conservative transfusion as a trigger of 24 with a target hematocrit of 24 to 27, whereas the liberal strategy was defined as a trigger of 30 with a target hematocrit of 30 to 33. This study showed that the liberal strategy was associated with a higher rate of the composite outcome of in-hospital death, recurrent MI, or new or worsening heart failure (38% vs. 13%; P = 0.046). Pooled data from the 2 studies showed no mortality benefit for aggressive compared with conservative RBC transfusion protocols (RR, 1.00 [CI, 0.68 to 1.46];  $I^2 = 0.0\%$ ).

3 December 2013 Annals of Internal Medicine Volume 159 • Number 11 771

#### Exercise Tolerance and Duration

Evidence was insufficient to determine the effect of RBC transfusions on exercise tolerance and duration.

#### Quality of Life

Evidence was insufficient to determine the effect of RBC transfusions on quality of life.

#### **Surgical Patients**

#### Mortality

Low-quality evidence from 3 studies (13, 14, 20) assessed short-term mortality in hip fracture and vascular surgery patients treated with liberal RBC transfusion (hemoglobin trigger, 10 g/dL) compared with restrictive transfusion (hemoglobin trigger, 8 to 9 g/dL) and found no difference in outcomes between the 2 treatments (RR, 1.35 [CI, 0.80 to 2.25];  $I^2 = 0.0\%$ ). Observational studies also failed to find a mortality benefit with aggressive transfusion (14, 21, 22).

#### Cardiovascular Events

Subgroup analysis in 1 study in vascular surgery patients found an increase in MI in patients transfused at a hemoglobin level of 9 g/dL or more compared with those transfused at hemoglobin levels ranging from 7 to 9 g/dL (21). Low-quality evidence from 2 studies (10, 15) did not find a statistically significant difference between liberal and restrictive RBC transfusion protocols in cardiovascular complications of MI (RR, 0.60 [CI, 0.34 to 1.03];  $I^2 = 0.0\%$ ).

#### Exercise Tolerance and Duration

Evidence was insufficient to determine the effect of RBC transfusions on exercise tolerance and duration.

#### Quality of Life

Evidence was insufficient to determine the effect of RBC transfusions on quality of life.

#### Observational Studies of RBC Transfusions in Different Populations of Patients With Heart Disease

Because few randomized, controlled trials assessed RBC transfusions for treatment of anemia in patients with heart disease, the evidence review also included observational studies in various patient populations. For more information on the descriptions, quality, and outcomes of the individual studies, refer to the background evidence review (8).

#### Percutaneous Coronary Intervention

Nine observational studies (23–33) evaluated blood transfusion in patients having percutaneous coronary intervention. The mean nadir hemoglobin levels across the studies were 8 to 9 g/dL. Overall, most studies showed that transfusion was associated with a higher mortality risk in the percutaneous coronary intervention population and suggested that this risk may be higher in nonbleeding anemic patients.

772 3 December 2013 Annals of Internal Medicine Volume 159 • Number 11

Twelve observational studies (25, 26, 32, 34–43) evaluated transfusion in patients with the acute coronary syndrome or MI. Overall, the data suggested that transfusion provides no benefit and may harm patients with heart disease and hemoglobin levels greater than 10 g/dL. Furthermore, patients with non–ST-segment elevation MI and hemoglobin levels ranging from 8 to 9 g/dL also do not have improved outcomes with transfusions.

#### Heart Failure

Evidence from 2 observational studies (44, 45) that assessed transfusion in patients with acute decompensated heart failure reported conflicting results on mortality.

#### HARMS OF TREATMENT OF ANEMIA WITH RBC TRANSFUSIONS

Adverse events potentially associated with RBC transfusions have been described in other patient populations and include CHF, fever, and transfusion-related acute lung injury. Reporting of harms for RBC transfusions for anemic patients with heart disease was sparse. No trials reported excess adverse events for liberal compared with restrictive transfusion.

#### BENEFITS OF TREATMENT OF ANEMIA WITH ESAS

Overall evidence from 16 randomized, controlled trials (9, 46-61) evaluating the effect of ESAs in patients with heart disease did not show that ESA use improved health outcomes (**Table 2**). The baseline hemoglobin levels for study patients ranged from 9 to 10 g/dL.

#### Mortality

High-quality evidence showed that ESA treatment did not improve mortality in anemic patients with stable CHF. Pooled data from 11 studies of patients with CHF or CHD (hemoglobin target levels, 12 to 15 g/dL) suggested an increased risk for mortality (RR, 1.07 [CI, 0.98 to 1.16];  $I^2 = 0.0\%$ ) for patients receiving ESA treatment compared with control patients (9, 46, 49, 50, 53–57, 60, 62).

#### Cardiovascular Events

High-quality evidence showed that ESAs do not affect cardiovascular events in patients with stable CHF. Pooled data from 7 studies (reported in 8 publications) (47, 48, 54–57, 60, 61) showed no difference in the risk for cardiovascular events when comparing ESA treatment with control (RR, 0.94 [CI, 0.82 to 1.08];  $I^2 = 41.5\%$ ). Hemoglobin target levels ranged from 9.0 to 15.0 g/dL in the studies.

#### **Exercise Tolerance and Duration**

Moderate-quality evidence showed that ESA treatment had no effect on exercise tolerance and duration in patients with stable CHF. Pooled data from 9 studies (46, 49, 50,

Treatment	Patient Population	Outcome	Effect*	Data	Quality of Evidence
RBC transfusions	Stable CHD (all patients)	Mortality	(~)	RR, 0.86 (95% CI, 0.46 to 1.62)	Low
		Cardiovascular events	(~)	RR, 0.60 (CI, 0.34 to 1.03)	Low
	ACS	Mortality	(~)	RR, 0.23 (CI, 0.05 to 1.02)	Low
		Cardiovascular events	(~)	RR, 0.70 (CI, 0.24 to 2.07)	Low
	Noncardiac surgery	Mortality	(~)	RR, 1.44 (CI, 0.81 to 2.54)	Low
		Cardiovascular events	(~)	RR, 0.52 (CI, 0.27 to 1.01)	Low
ESAs	Stable CHF	Mortality	(~)	RR, 1.07 (CI, 0.98 to 1.16)	High
		Cardiovascular events	(~)	RR, 0.94 (CI, 0.82 to 1.08)	High
		Quality of life	(~)	No consistent differences	Moderate
		Exercise tolerance and duration	(~)	Mean difference in NYHA, -0.77 (CI, -1.21 to -0.32)	Moderate
		Hospitalizations	(~)	RR, 0.97 (CI, 0.87 to 1.10)	High
		Harms, including hypertension and cerebrovascular and thrombotic events	(-)	Hypertension: RR, 1.20 (CI, 0.90 to 1.59) Ischemic stroke: RR, 1.33 (CI, 0.93 to 1.89) VTE: RR, 1.36 (CI, 1.17 to 1.58)	Moderate
	Stable CHD†	Mortality	(-)	Increased mortality	Low
		Cardiovascular events	(~)	No effect	Low
		Quality of life	(~)	No effect	Low
		Venous thrombosis	(-)	Increased risk for venous thrombosis	Low
IV iron	Stable CHF	Mortality	(~)	NA	Insufficient
		Cardiovascular events	(+)	27.6% vs. 50.2% (P = 0.01)	Low
		Exercise tolerance and duration	(+)	Improvements in NYHA class and 6-min walk test	Moderate
		Quality of life	(+)	Improvement on various scales	Moderate
		Serious harms	(~)	No differences, although harms were sparsely reported	Moderate

#### Table 2. Evidence for the Effects of RBC Transfusions. ESAs and IV Iron for the Treatment of Anemia in Patients With Heart Disease

ACS = acute coronary syndrome; CHD = coronary heart disease; CHF = congestive heart failure; ESA = erythropoiesis-stimulating agent; IV = intravenous; NA = not \*Effect: (+) indicates that treatment provided benefit; (-) indicates that treatment resulted in harm; ( $\sim$ ) indicates mixed findings or no effect.

+ Patients included in these studies had advanced kidney disease and/or end-stage renal disease, so the application to other patient populations is unclear.

52, 53, 55-57, 59) showed that treatment with ESAs in patients with CHF (hemoglobin target levels, 12.0 to 15.0 g/dL) resulted in improved NYHA functional class scores compared with control patients (mean difference, -0.77 $[CI, -1.12 \text{ to } -0.32]; I^2 = 96\%$ ). However, the results were generally inconsistent and the studies were highly heterogeneous. Limiting the analysis to the 4 methodologically rigorous studies (50, 53, 56, 57) showed no statistically significant difference in NYHA scores (NYHA score, -0.15 [CI, -0.36 to 0.06];  $I^2 = 62.1\%$ ). Pooled data from 4 studies (51, 52, 56, 58) (hemoglobin target levels, 12.5 to 15.0 g/dL) showed that ESA treatment resulted in a nonstatistically significant increase in the 6-minute walk test, and the studies were heterogeneous (mean change in distance walked, 74.4 m [CI, -0.16 to 149.0 m];  $I^2 =$ 88.7%). Reported clinically important differences in the 6-minute walk test range from 43 to 54 m (63). Two studies used the Naughton protocol to assess change in exercise treadmill time. The smaller trial showed a slight increase in exercise duration with ESA treatment (62); however, the larger trial showed no difference (53).

#### Quality of Life

Moderate-quality evidence from 6 studies (51, 53, 56-58) showed that treatment with ESAs did not improve quality of life in anemic patients with stable CHF (hemoglobin target levels, 13.0 to 15.0 g/dL in the studies). The variability of tools used to assess quality of life and incon-

www.annals.org

sistencies in the results limited analysis of this outcome. Two of the 4 trials (53, 56-58) that used the Patient Global Assessment scale showed that ESA treatment improved scores. One trial (58) had methodological flaws, and 1 study (57) found no significant differences in the Kansas City Cardiomyopathy Questionnaire and Minnesota Living With Heart Failure Questionnaire scores. One of the 4 trials (53, 56-58) that evaluated the Minnesota Living With Heart Failure Questionnaire scores showed improvement with treatment, although this study had a high risk of bias (58). Of the 3 studies (51, 56, 57) that evaluated patients by using the Kansas City Cardiomyopathy Questionnaire, 1 study reported an improvement from baseline "total symptom score" (56), whereas another study reported improvements in "functional score" and "summary score" (51). Low-quality evidence from 1 study of patients with CHD and end-stage renal disease showed no improvement in quality of life (60).

#### Hospitalization

High-quality evidence showed that hospitalizations were not statistically significantly different for patients with stable CHF. Limiting the analysis to the 3 trials with a low risk of bias (53, 54, 57, 60) showed no difference in hospitalizations (RR, 0.97 [CI, 0.87 to 1.10];  $I^2 = 0.0\%$ ). Pooling data from all 8 studies showed that ESA treatment was associated with a reduction in hospitalizations compared with control (RR, 0.69 [CI, 0.52 to 0.93];  $I^2 =$ 

3 December 2013 Annals of Internal Medicine Volume 159 • Number 11 773

37.7%); however, pooling data from only the larger and higher-quality studies showed no effect. Hemoglobin target levels in the studies ranged from 13.0 to 15.0 g/dL.

# HARMS OF TREATMENT OF ANEMIA WITH ESAS Hypertension

Moderate-quality evidence pooled from 7 studies of patients with CHF (48, 50–53, 56, 57) showed that ESA treatment was associated with a nonstatistically significant increased risk for hypertension compared with control (RR, 1.20 [CI, 0.90 to 1.59];  $I^2 = 0.0\%$ ) (hemoglobin target levels, 9.0 to 15.0 g/dL).

#### **Cerebrovascular Events**

Moderate-quality evidence from 4 studies (51–53, 57) reported on cerebrovascular events in patients with CHF. However, few events were reported, and no difference between ESA and control groups was shown (RR, 1.33 [CI, 0.93 to 1.89];  $I^2 = 15.9\%$ ) (hemoglobin target levels, 12.5 to 14.0 g/dL).

#### Venous Thrombosis

Moderate-quality evidence from 4 studies in patients with CHF showed that ESAs (hemoglobin target levels, 12.5 to 15.0 g/dL) increased the risk for venous thrombosis, with 1 trial reporting on patients with chronic kidney disease and diabetes (RR, 1.36 [CI, 1.17 to 1.58]). A recent randomized, double-blind trial of patients with systolic heart failure and anemia found a significant increase in thromboembolic events in those treated with darbepoetin- $\alpha$  to a goal hemoglobin level greater than 13 mg/dL (9).

# INFLUENCE OF HEMOGLOBIN TARGET LEVELS ON OUTCOMES

No studies assessed the effects of moderate compared with lower hemoglobin target levels on outcomes in patients with heart disease. All of the small trials comparing ESAs with placebo in patients with heart failure included patients with moderate anemia and a mean baseline hemoglobin level within the narrow range (10 to 12 g/dL). In all case patients, ESA use was associated with a statistically significant increase in hemoglobin level (mean range, 1.6 to 2.8 g/dL).

Three studies (47, 48, 54) evaluated the titration of ESAs to target normal hemoglobin levels compared with lower targets (9 to 11.3 g/dL) in patients with comorbid chronic kidney disease and heart disease and found no benefit from aggressive ESA use. Two of the studies (48, 54) showed that aggressive ESA use to normalize hemoglobin level increased the risk for venous thromboembolic events and suggested increased mortality rates (48, 54).

774 3 December 2013 Annals of Internal Medicine Volume 159 • Number 11

#### BENEFITS OF USING INTRAVENOUS IRON TO TREAT IRON DEFICIENCY WITH OR WITHOUT ANEMIA

Three studies (64-66) assessed the effect of intravenous (IV) iron in patients with heart failure. Data were primarily derived from 1 large study, which included patients with and without anemia and found similar results for both groups (64) (Table 2).

#### Mortality

Evidence was insufficient to determine the effect of IV iron treatment on mortality.

#### **Cardiovascular Events**

Low-quality evidence from 1 study (64) found that 27.6% of patients treated with IV iron carboxymaltose had cardiovascular events compared with 50.2% of patients receiving placebo (P = 0.01). However, this end point was not prespecified in the study, and the definition of the outcome was unclear.

#### Exercise Tolerance and Duration

Moderate-quality evidence showed that IV iron administration increased exercise tolerance and duration in patients with stable CHF and chronic kidney disease no worse than stage 3. The largest trial, FAIR-HF (Ferinject Assessment in Patients With Iron Deficiency and Chronic Heart Failure) included anemic and nonanemic patients, with most patients having ferritin levels less than 100  $\mu$ g/L (64). This trial showed that 200 mg of IV ferric carboxymaltose increased 6-minute walk distance (313 m vs. 277 m) compared with IV saline (64). A second study found that among patients with iron deficiency, anemia, chronic heart failure, and chronic kidney disease, treatment with 200 mg of IV iron sucrose weekly for 5 weeks increased 6-minute walk distance compared with saline (66). The FERRIC-HF (Ferric Iron Sucrose in Heart Failure) (65) study showed that patients who received 200 mg of IV iron sucrose had improved exercise duration compared with placebo. However, this trial had a high risk of bias due to lack of patient blinding.

#### Quality of Life

Moderate-quality evidence showed that IV iron improved quality of life in patients with anemia or iron deficiency, stable CHF, and chronic kidney disease no worse than stage 3. The FAIR-HF study showed that IV iron treatment improved Patient Global Assessment scores compared with control patients (50% vs. 28%; odds ratio, 2.51 [CI, 1.75 to 3.61]) and improved NYHA functional class (odds ratio for improvement by 1 class, 2.40 [CI, 1.55 to 3.71]), regardless of anemia status (hemoglobin level  $\leq 12$  g/dL) (64). This trial also showed improved quality-of-life scores as assessed by the European Quality of Life–5 Dimensions (EQ-5D) (63 vs. 67) (64). Two other studies showed that patients who received 200 mg of IV iron sucrose had improved Minnesota Living With Heart Failure Questionnaire and NYHA scores (65, 66).

#### Harms of Using Intravenous Iron to Treat Iron Deficiency With or Without Anemia

Moderate-quality evidence from 3 studies (64–66) found no statistically significant difference in serious harms between IV iron treatment and control. However, harms were sparsely reported, and there is no available evidence on long-term outcomes.

#### SUMMARY

Management of patients with heart disease and anemia might appropriately differ from that of the general population. Hence, understanding the evidence base and using clinical judgment are critical when managing these patients. Anemia is associated with worse outcomes in patients with CHF or CHD. However, it is uncertain if anemia is the cause of poorer outcomes or if it is a marker of poor outcomes and reflects advanced cardiovascular disease. See **Table 2** for a summary of the evidence reviewed in this guideline.

Low-quality evidence found that RBC transfusion using restrictive compared with liberal transfusion protocols had no effect on mortality in patients with CHD. Observational studies suggested that transfusion is not beneficial and may be harmful among patients with heart disease and hemoglobin levels greater than 10 g/dL. A recently published trial indicated a trend toward improved cardiovascular outcomes in patients with anemia and unstable or stable CHD (10). However, differences in the baseline characteristics of the study groups and the small size of this pilot study do not answer the key clinical questions above, even after inclusion in the meta-analysis.

Overall, moderate-quality evidence showed no benefit from ESAs for improving exercise tolerance and duration or quality of life, and high-quality evidence showed no mortality benefit. Serious harms associated with the treatment include mortality and vascular thrombosis. A recently published trial showed no benefit across all subgroups studied and showed increased harms among those treated to a goal hemoglobin level greater than 13 g/dL (9). The evidence for ESA use is most applicable to patients with CHF and systolic dysfunction.

Few studies addressed IV iron therapy for patients with heart disease, and data on long-term harms were unavailable. One good-quality study among patients with chronic stable systolic heart failure showed that IV iron carboxymaltose improved exercise tolerance, quality of life, and exercise duration. Overall, moderate-quality evidence showed that IV iron therapy reduced cardiovascular events, and low-quality evidence found a mortality benefit. The evidence for IV iron therapy is most applicable to patients with NYHA class III heart failure and low ferritin levels. Refer to the **Figure** for a summary of the recommendations and clinical considerations.

Recommendation 1: ACP recommends using a restrictive red blood cell transfusion strategy (trigger hemoglobin threshold of 7 to 8 g/dL compared with higher hemoglobin levels) in

www.annals.org

# hospitalized patients with coronary heart disease. (Grade: weak recommendation; low-quality evidence)

Low-quality evidence showed no mortality benefit of liberal compared with restrictive transfusion strategies across the patient populations studied. Most evidence did not show a substantial difference in benefit between liberal and restrictive RBC transfusions. In addition, low-quality evidence showed conflicting results for cardiovascular events. Low-quality evidence showed no mortality benefit of a higher (liberal) transfusion threshold (hemoglobin levels >10 g/dL) and fewer cardiovascular events with a lower (restrictive) transfusion threshold (hemoglobin levels of 7 to 8 g/dL) in noncardiac surgery patients. Studies showed no short-term mortality benefit in hip fracture and vascular surgery patients treated with liberal RBC transfusion compared with restrictive transfusion and found no difference in outcomes. Observational studies also failed to find a mortality benefit with aggressive liberal transfusion. For noncardiac surgeries other than hip fracture surgery, data are inconclusive. Low-quality evidence showed that patients with the acute coronary syndrome who received liberal RBC transfusions (hemoglobin threshold of 10 g/dL) had a mortality benefit compared with those who received more restrictive transfusion thresholds. However, this result was from a small study that included patients with stable and unstable CHD, and the difference was not statistically significant. Harms were sparsely reported, and no trials reported a difference in adverse events for liberal compared with restrictive transfusions.

Recommendation 2: ACP recommends against the use of erythropoiesis-stimulating agents in patients with mild to moderate anemia and congestive heart failure or coronary heart disease. (Grade: strong recommendation; moderatequality evidence)

The harms outweigh the benefits for treating patients with mild to moderate anemia using ESAs. The potential harms associated with ESA therapy include increased risk for thromboembolic events shown in 3 studies and a suggestion of increased stroke rates in 1 study. Although anemia is common in patients with CHF and CHD, highquality evidence showed that treatment with ESAs did not improve mortality or affect cardiovascular events or hospitalizations, and moderate-quality evidence showed no improvement for quality of life. Baseline hemoglobin levels for study patients ranged from 9 to 10 g/dL.

#### INCONCLUSIVE AREAS OF EVIDENCE

Emerging evidence shows a short-term benefit from IV iron carboxymaltose in patients with CHF and ferritin levels less than 100  $\mu$ g/L. However, evidence is lacking on long-term outcomes, and evidence on harms was sparsely reported. The effect of oral administration of iron and how it compares with IV iron for treating anemic patients with heart disease is unknown. Current evidence suggests that IV iron treatment improves exercise tolerance and quality

<sup>3</sup> December 2013 Annals of Internal Medicine Volume 159 • Number 11 775

Figure. Summary of the American College of Physicians guideline on treatment of anemia in patients with heart disease.



#### Summary of the American College of Physicians Guideline on Treatment of Anemia in Patients With Heart Disease

Disease/Condition	Anemia and heart disease		
Target Audience	Internists, family physicians, and other clinicians		
Target Patient Population	Adult patients with symptomatic CHF (with or without reduced systolic function) or CHD (acute coronary syndrome, postacute coronary syndrome, or a history of MI or angina) and anemia or iron deficiency		
Interventions	Red blood cell transfusion, ESAs with or without iron (including erythropoietin and darbepoetin), and intravenous iron		
Outcomes	Mortality (all-cause and disease-specific); cardiovascular events (MI, CHF exacerbation, arrhythmia, or cardiac death); exercise tolerance (any metric, most commonly NYHA class, 6-min walk test); quality of life; hospitalization (all-cause and disease-specific); and harms, including hypertension, venous thromboembolic events, and ischemic cerebrovascular events		
Benefits of Treatment	Red blood cell transfusion: no benefits shown when comparing liberal to restrictive transfusion		
	ESAs: no benefit		
	Intravenous iron: increased exercise tolerance, improved quality of life		
Harms of Treatment	Red blood cell transfusion: adverse events potentially associated with red blood cell transfusions, such as fever, transfusion- related acute lung injury, and congestive heart failure		
	ESAs: hypertension and venous thrombosis		
	Intravenous iron: no harms identified but sparsely reported		
Recommendations	Recommendation 1: ACP recommends using a restrictive red blood cell transfusion strategy (trigger hemoglobin threshold of 7–8 g/dL compared with higher hemoglobin levels) in hospitalized patients with coronary heart disease. (Grade: weak recommendation; low-quality evidence)		
	Recommendation 2: ACP recommends against the use of erythropoiesis-stimulating agents in patients with mild to moderate anemia and congestive heart failure or coronary heart disease. (Grade: strong recommendation; moderate-quality evidence)		
High Value Care	Current evidence does not support the benefit of liberal blood transfusions in patients with asymptomatic anemia and heart disease. Therefore, the ACP does not support the liberal use of blood transfusions in the management of mild to moderate anemia in patients with cardiovascular disease. The probability that transfusion may be beneficial is higher in patients with lower hemoglobin levels (<7 g/dL) and lower in less anemic patients (hemoglobin >10 g/dL) (67). The ACP does not support the use of ESAs for treating patients with mild to moderate anemia and heart disease because the harms outweigh the benefits for these patients.		
Clinical Considerations	Patients with heart disease may have anemia because of iron deficiency, use of ACE inhibitors, renal insufficiency, and poor nutrition.		
	Presence of anemia is associated with increased mortality and morbidity. However, it is uncertain if anemia is an independent risk factor for poor outcomes or if it is a marker of more severe illness.		
	The impact of oral administration of iron and how it compares with IV iron for treating anemic patients with heart disease is unknown.		

ACE = angiotensin-converting enzyme; CHD = coronary heart disease; CHF = congestive heart failure; ESA = erythropoiesis-stimulating agent; MI = myocardial infarction; NYHA = New York Heart Association.

of life and reduces mortality and hospitalizations. Although the evidence is intriguing and positive, it is limited to only 3 studies at this time, and the data were primarily derived from 1 large trial that included patients with and without anemia. Note that at the time of writing of this guideline, ferrous carboxymaltose was not yet approved for IV treatment of anemic patients in the United States.

#### ACP HIGH-VALUE CARE

Current evidence does not support the benefit of liberal blood transfusions in patients with asymptomatic anemia and heart disease. Therefore, ACP does not support this practice for management of mild to moderate anemia in patients with cardiovascular disease. The probability

776 3 December 2013 Annals of Internal Medicine Volume 159 • Number 11

Downloaded From: http://annals.org/ on 10/14/2014

that transfusion may be beneficial is greater in patients with lower hemoglobin levels (<7 g/dL) and lower in less anemic patients (hemoglobin levels >10 g/dL) (67). The ACP does not support use of ESAs in cases of mild to moderate anemia and heart disease because the harms outweigh the benefits for these patients.

From the American College of Physicians, Philadelphia, Pennsylvania; Oregon Health and Science University, Portland, Oregon; Huntington Hospital, Pasadena, California; and West Los Angeles Veteran Affairs Medical Center, Los Angeles, California.

**Note:** Clinical practice guidelines are "guides" only and may not apply to all patients and clinical situations. Thus, they are not intended to override clinicians' judgment. All ACP clinical practice guidelines are con-

sidered automatically withdrawn or invalid 5 years after publication or once an update has been issued.

**Disclaimer:** The authors of this article are responsible for its contents, including any clinical or treatment recommendations. No statement in this article should be construed as an official position of the U.S. Department of Veterans Affairs.

**Financial Support:** Financial support for the development of this guideline comes exclusively from the ACP operating budget.

**Potential Conflicts of Interest:** Dr. Shekelle: *Grants:* Agency for Healthcare Research and Quality, Veterans Affairs, Centers for Medicare & Medicaid Services, Office of the National Coordinator for Health Information Technology; *Personal fees:* ECRI Institute, Veterans Affairs, UpToDate. All other authors have no disclosures. Disclosures can also be viewed at www.acponline.org/authors/icmje/ConflictOfInterestForms.do?msNum=M13-1830. A record of conflicts of interest is kept for each Clinical Guidelines Committee meeting and conference call and can be viewed at www.acponline.org/clinical\_information/guidelines/guidelines/conflicts\_cgc.htm.

Requests for Single Reprints: Amir Qaseem, MD, PhD, MHA, American College of Physicians, 190 N. Independence Mall West, Philadelphia, PA 19106; e-mail, aqaseem@acponline.org.

Current author addresses and author contributions are available at www .annals.org.

#### References

1. Boyd CM, Leff B, Wolff JL, Yu Q, Zhou J, Rand C, et al. Informing clinical practice guideline development and implementation: prevalence of coexisting conditions among adults with coronary heart disease. J Am Geriatr Soc. 2011; 59:797-805. [PMID: 21568950]

2. Felker GM, Adams KF Jr, Gattis WA, O'Connor CM. Anemia as a risk factor and therapeutic target in heart failure. J Am Coll Cardiol. 2004;44:959-66. [PMID: 15337204]

3. Malyszko J, Bachorzewska-Gajewska H, Malyszko J, Levin-Iaina N, Iaina A, Dobrzycki S. Prevalence of chronic kidney disease and anemia in patients with coronary artery disease with normal serum creatinine undergoing percutaneous coronary interventions: relation to New York Heart Association class. Isr Med Assoc J. 2010;12:489-93. [PMID: 21337818]

4. Komajda M, Anker SD, Charlesworth A, Okonko D, Metra M, Di Lenarda A, et al. The impact of new onset anaemia on morbidity and mortality in chronic heart failure: results from COMET. Eur Heart J. 2006;27:1440-6. [PMID: 16717081]

5. Kosiborod M, Smith GL, Radford MJ, Foody JM, Krumholz HM. The prognostic importance of anemia in patients with heart failure. Am J Med. 2003; 114:112-9. [PMID: 12586230]

6. Silverberg DS, Wexler D, Blum M, Keren G, Sheps D, Leibovitch E, et al. The use of subcutaneous erythropoietin and intravenous iron for the treatment of the anemia of severe, resistant congestive heart failure improves cardiac and renal function and functional cardiac class, and markedly reduces hospitalizations. J Am Coll Cardiol. 2000;35:1737-44. [PMID: 10841219]

7. Kansagara D, Dyer E, Englander H, Fu R, Freeman M, Kagen D. Treatment of anemia in patients with heart disease. A systematic review. Ann Intern Med. 2013;159:746-57.

8. Kansagara D, Dyer E, Englander H, Freeman M, Kagen D. Treatment of Anemia in Patients with Heart Disease: A Systematic Review. VA-ESP Project no. 05-225. 2011. Accessed at www.hsrd.research.va.gov/publications/esp /anemia.cfm on 18 October 2013.

9. Swedberg K, Young JB, Anand IS, Cheng S, Desai AS, Diaz R, et al; RED-HF Committees. Treatment of anemia with darbepoetin alfa in systolic heart failure. N Engl J Med. 2013;368:1210-9. [PMID: 23473338]

10. Carson JL, Brooks MM, Abbott JD, Chaitman B, Kelsey SF, Triulzi DJ, et al. Liberal versus restrictive transfusion thresholds for patients with symptom-

www.annals.org

atic coronary artery disease. Am Heart J. 2013;165:964-971.e1. [PMID: 23708168]

11. Higgins JPT, Altman DG, eds. Chapter 8: Assessing risk of bias in included studies. In: Higgins JPT, Green S, eds. Cochrane Handbook for Systematic Reviews of Interventions. The Cochrane Collaboration; 2008, version 5.0.1. Accessed at www.cochrane-handbook.org on 17 October 2013.

12. Qaseem A, Snow V, Owens DK, Shekelle P; Clinical Guidelines Committee of the American College of Physicians. The development of clinical practice guidelines and guidance statements of the American College of Physicians: summary of methods. Ann Intern Med. 2010;153:194-9. [PMID: 20679562]

13. Bush RL, Pevec WC, Holcroft JW. A prospective, randomized trial limiting perioperative red blood cell transfusions in vascular patients. Am J Surg. 1997; 174:143-8. [PMID: 9293831]

14. Carson JL, Terrin ML, Barton FB, Aaron R, Greenburg AG, Heck DA, et al. A pilot randomized trial comparing symptomatic vs. hemoglobin-leveldriven red blood cell transfusions following hip fracture. Transfusion. 1998;38: 522-9. [PMID: 9661685]

15. Carson JL, Terrin ML, Noveck H, Sanders DW, Chaitman BR, Rhoads GG, et al; FOCUS Investigators. Liberal or restrictive transfusion in high-risk patients after hip surgery. N Engl J Med. 2011;365:2453-62. [PMID: 22168590]

16. Hébert PC, Yetisir E, Martin C, Blajchman MA, Wells G, Marshall J, et al; Transfusion Requirements in Critical Care Investigators for the Canadian Critical Care Trials Group. Is a low transfusion threshold safe in critically ill patients with cardiovascular diseases? Crit Care Med. 2001;29:227-34. [PMID: 11246298]

17. Cooper HA, Rao SV, Greenberg MD, Rumsey MP, McKenzie M, Alcorn KW, et al. Conservative versus liberal red cell transfusion in acute myocardial infarction (the CRIT Randomized Pilot Study). Am J Cardiol. 2011;108:1108-11. [PMID: 21791325]

18. Cooper HA, Rao SV, Greenberg MD, Rumsey MP, McKenzie M, Alcorn KW, et al. Conservative versus liberal red cell transfusion in acute myocardial infarction (the CRIT Randomized Pilot Study). Am J Cardiol. 2011;108:1108-11. [PMID: 21791325]

19. Hébert PC, Wells G, Blajchman MA, Marshall J, Martin C, Pagliarello G, et al. A multicenter, randomized, controlled clinical trial of transfusion requirements in critical care. Transfusion Requirements in Critical Care Investigators, Canadian Critical Care Trials Group. N Engl J Med. 1999;340:409-17. [PMID: 9971864]

20. Carson J, Terrin M, Magaziner J, Sanders D, Cook D, Hildebrand K. Transfusion trigger trial for functional outcomes in cardiovascular patients undergoing surgical hip fracture repair (FOCUS): the principle results. Blood. 2009; 114.

21. Bursi F, Barbieri A, Politi L, Di Girolamo A, Malagoli A, Grimaldi T, et al. Perioperative red blood cell transfusion and outcome in stable patients after elective major vascular surgery. Eur J Vasc Endovasc Surg. 2009;37:311-8. [PMID: 19111480]

22. Glance LG, Dick AW, Mukamel DB, Fleming FJ, Zollo RA, Wissler R, et al. Association between intraoperative blood transfusion and mortality and morbidity in patients undergoing noncardiac surgery. Anesthesiology. 2011;114: 283-92. [PMID: 21239971]

23. Chase AJ, Fretz EB, Warburton WP, Klinke WP, Carere RG, Pi D, et al. Association of the arterial access site at angioplasty with transfusion and mortality: the M.O.R.T.A.L study (Mortality benefit Of Reduced Transfusion after percutaneous coronary intervention via the Arm or Leg). Heart. 2008;94:1019-25. [PMID: 18332059]

24. Doyle BJ, Ting HH, Bell MR, Lennon RJ, Mathew V, Singh M, et al. Major femoral bleeding complications after percutaneous coronary intervention: incidence, predictors, and impact on long-term survival among 17,901 patients treated at the Mayo Clinic from 1994 to 2005. JACC Cardiovasc Interv. 2008; 1:202-9. [PMID: 19463301]

25. Jani SM, Smith DE, Share D, Kline-Rogers E, Khanal S, O'Donnell MJ, et al. Blood transfusion and in-hospital outcomes in anemic patients with myocardial infarction undergoing percutaneous coronary intervention. Clin Cardiol. 2007;30:II49-56. [PMID: 18228652]

26. Jolicoeur EM, O'Neill WW, Hellkamp A, Hamm CW, Holmes DR Jr, Al-Khalidi HR, et al; APEX-AMI Investigators. Transfusion and mortality in patients with ST-segment elevation myocardial infarction treated with primary percutaneous coronary intervention. Eur Heart J. 2009;30:2575-83. [PMID: 19596659]

3 December 2013 Annals of Internal Medicine Volume 159 • Number 11 777

#### CLINICAL GUIDELINE | Treating Anemia in Patients With Heart Disease

27. Kim P, Dixon S, Eisenbrey AB, O'Malley B, Boura J, O'Neill W. Impact of acute blood loss anemia and red blood cell transfusion on mortality after percutaneous coronary intervention. Clin Cardiol. 2007;30:II35-43. [PMID: 18228650]

28. Kinnaird TD, Stabile E, Mintz GS, Lee CW, Canos DA, Gevorkian N, et al. Incidence, predictors, and prognostic implications of bleeding and blood transfusion following percutaneous coronary interventions. Am J Cardiol. 2003; 92:930-5. [PMID: 14556868]

29. Maluenda G, Lemesle G, Ben-Dor I, Collins SD, Syed AI, Li Y, et al. Value of blood transfusion in patients with a blood hematocrit of 24% to 30% after percutaneous coronary intervention. Am J Cardiol. 2009;104:1069-73. [PMID: 19801026]

30. Maluenda G, Lemesle G, Syed A, Collins SD, Ben-Dor I, Li Y, et al. Should patients who develop anemia (hematocrit (less-than or equal to) 27%) after percutaneous coronary intervention be transfused? JACC. 2009;53:A84.

31. Maluenda G, Lemesle G, Syed A, Collins SD, Ben-Dor I, Li Y, et al. Does transfusion for major bleeding after percutaneous coronary intervention impact clinical outcome in patients admitted with normal Hematocrit? JACC. 2009;53: A72.

32. Nikolsky E, Mehran R, Sadeghi HM, Grines CL, Cox DA, Garcia E, et al. Prognostic impact of blood transfusion after primary angioplasty for acute myocardial infarction: analysis from the CADILLAC (Controlled Abciximab and Device Investigation to Lower Late Angioplasty Complications) Trial. JACC Cardiovasc Interv. 2009;2:624-32. [PMID: 19628185]

33. Yatskar L, Selzer F, Feit F, Cohen HA, Jacobs AK, Williams DO, et al. Access site hematoma requiring blood transfusion predicts mortality in patients undergoing percutaneous coronary intervention: data from the National Heart, Lung, and Blood Institute Dynamic Registry. Catheter Cardiovasc Interv. 2007; 69:961-6. [PMID: 17421023]

34. Aggarwal C, Panza JA, Cooper HA. Does the relation between red blood cell transfusion and mortality vary according to transfusion indication? A case-control study among patients with acute coronary syndromes. Coron Artery Dis. 2011; 22:194-8. [PMID: 21169814]

35. Alexander KP, Chen AY, Wang TY, Rao SV, Newby LK, LaPointe NM, et al; CRUSADE Investigators. Transfusion practice and outcomes in non-ST-segment elevation acute coronary syndromes. Am Heart J. 2008;155:1047-53. [PMID: 18513518]

36. Aronson D, Dann EJ, Bonstein L, Blich M, Kapeliovich M, Beyar R, et al. Impact of red blood cell transfusion on clinical outcomes in patients with acute myocardial infarction. Am J Cardiol. 2008;102:115-9. [PMID: 18602505]

Rao SV, Jollis JG, Harrington RA, Granger CB, Newby LK, Armstrong PW, et al. Relationship of blood transfusion and clinical outcomes in patients with acute coronary syndromes. JAMA. 2004;292:1555-62. [PMID: 15467057]
Sabatine MS, Morrow DA, Giugliano RP, Burton PB, Murphy SA, McCabe CH, et al. Association of hemoglobin levels with clinical outcomes in acute coronary syndromes. Circulation. 2005;111:2042-9. [PMID: 15824203]
Shishehbor MH, Madhwal S, Rajagopal V, Hsu A, Kelly P, Gurm HS, et al. Impact of blood transfusion on short- and long-term mortality in patients with ST-segment elevation myocardial infarction. JACC Cardiovasc Interv. 2009; 2:46-53. [PMID: 19463397]

40. Singla I, Zahid M, Good CB, Macioce A, Sonel AF. Impact of blood transfusions in patients presenting with anemia and suspected acute coronary syndrome. Am J Cardiol. 2007;99:1119-21. [PMID: 17437739]

41. Wu WC, Rathore SS, Wang Y, Radford MJ, Krumholz HM. Blood transfusion in elderly patients with acute myocardial infarction. N Engl J Med. 2001; 345:1230-6. [PMID: 11680442]

42. Yang X, Alexander KP, Chen AY, Roe MT, Brindis RG, Rao SV, et al; CRUSADE Investigators. The implications of blood transfusions for patients with non-ST-segment elevation acute coronary syndromes: results from the CRUSADE National Quality Improvement Initiative. J Am Coll Cardiol. 2005; 46:1490-5. [PMID: 16226173]

43. Athar MK, Bagga S, Nair N, Punjabi V, Vito K, Schorr C, et al. Risk of cardiac arrhythmias and conduction abnormalities in patients with acute myocardial infarction receiving packed red blood cell transfusions. J Crit Care. 2011;26: 335-41. [PMID: 20869199]

44. Garty M, Cohen E, Zuchenko A, Behar S, Boyko V, Iakobishvili Z, et al; Heart Failure Survey in ISrael (HFSIS) Investigators. Blood transfusion for acute decompensated heart failure—friend or foe? Am Heart J. 2009;158:653-8. [PMID: 19781427]

778 3 December 2013 Annals of Internal Medicine Volume 159 • Number 11

45. Kao DP, Kreso E, Fonarow GC, Krantz MJ. Characteristics and outcomes among heart failure patients with anemia and renal insufficiency with and without blood transfusions (public discharge data from California 2000-2006). Am J Cardiol. 2011;107:69-73. [PMID: 21146689]

46. Comín-Colet J, Ruiz S, Cladellas M, Rizzo M, Torres A, Bruguera J. A pilot evaluation of the long-term effect of combined therapy with intravenous iron sucrose and erythropoietin in elderly patients with advanced chronic heart failure and cardio-renal anemia syndrome: influence on neurohormonal activation and clinical outcomes. J Card Fail. 2009;15:727-35. [PMID: 19879457]

47. Szczech LA, Barnhart HX, Sapp S, Felker GM, Hernandez A, Reddan D, et al. A secondary analysis of the CHOIR trial shows that comorbid conditions differentially affect outcomes during anemia treatment. Kidney Int. 2010;77:239-46. [PMID: 19890274]

48. Pfeffer MA, Burdmann EA, Chen CY, Cooper ME, de Zeeuw D, Eckardt KU, et al; TREAT Investigators. Baseline characteristics in the Trial to Reduce Cardiovascular Events With Aranesp Therapy (TREAT). Am J Kidney Dis. 2009;54:59-69. [PMID: 19501439]

49. Palazzuoli A, Silverberg DS, Calabrò A, Spinelli T, Quatrini I, Campagna MS, et al. Beta-erythropoietin effects on ventricular remodeling, left and right systolic function, pulmonary pressure, and hospitalizations in patients affected with heart failure and anemia. J Cardiovasc Pharmacol. 2009;53:462-7. [PMID: 19455052]

50. Parissis JT, Kourea K, Andreadou I, Ikonomidis I, Markantonis S, Ioannidis K, et al. Effects of darbepoetin alfa on plasma mediators of oxidative and nitrosative stress in anemic patients with chronic heart failure secondary to ischemic or idiopathic dilated cardiomyopathy. Am J Cardiol. 2009;103:1134-8. [PMID: 19361602]

51. Kourea K, Parissis JT, Farmakis D, Paraskevaidis I, Panou F, Filippatos G, et al. Effects of darbepoetin-alpha on quality of life and emotional stress in anemic patients with chronic heart failure. Eur J Cardiovasc Prev Rehabil. 2008; 15:365-9. [PMID: 18525396]

52. Parissis JT, Kourea K, Panou F, Farmakis D, Paraskevaidis I, Ikonomidis I, et al. Effects of darbepoetin alpha on right and left ventricular systolic and diastolic function in anemic patients with chronic heart failure secondary to ischemic or idiopathic dilated cardiomyopathy. Am Heart J. 2008;155:751.e1-7. [PMID: 18371487]

53. Ghali JK, Anand IS, Abraham WT, Fonarow GC, Greenberg B, Krum H, et al; Study of Anemia in Heart Failure Trial (STAMINA-HeFT) Group. Randomized double-blind trial of darbepoetin alfa in patients with symptomatic heart failure and anemia. Circulation. 2008;117:526-35. [PMID: 18195176]

54. Besarab A, Goodkin DA, Nissenson AR; Normal Hematocrit Cardiac Trial Authors. The normal hematocrit study—follow-up [Letter]. N Engl J Med. 2008;358:433-4. [PMID: 18216370]

55. Palazzuoli A, Silverberg DS, Iovine F, Calabrò A, Campagna MS, Gallotta M, et al. Effects of beta-erythropoietin treatment on left ventricular remodeling, systolic function, and B-type natriuretic peptide levels in patients with the cardiorenal anemia syndrome. Am Heart J. 2007;154:645.e9-15. [PMID: 17892986]

56. van Veldhuisen DJ, Dickstein K, Cohen-Solal A, Lok DJ, Wasserman SM, Baker N, et al. Randomized, double-blind, placebo-controlled study to evaluate the effect of two dosing regimens of darbepoetin alfa in patients with heart failure and anaemia. Eur Heart J. 2007;28:2208-16. [PMID: 17681958]

57. Ponikowski P, Anker SD, Szachniewicz J, Okonko D, Ledwidge M, Zymlinski R, et al. Effect of darbepoetin alfa on exercise tolerance in anemic patients with symptomatic chronic heart failure: a randomized, double-blind, placebocontrolled trial. J Am Coll Cardiol. 2007;49:753-62. [PMID: 17306703]

58. Mancini DM, Kunavarapu C. Effect of erythropoietin on exercise capacity in anemic patients with advanced heart failure. Kidney Int Suppl. 2003:S48-52. [PMID: 14531773]

59. Silverberg DS, Wexler D, Sheps D, Blum M, Keren G, Baruch R, et al. The effect of correction of mild anemia in severe, resistant congestive heart failure using subcutaneous erythropoietin and intravenous iron: a randomized controlled study. J Am Coll Cardiol. 2001;37:1775-80. [PMID: 11401110]

60. Besarab A, Bolton WK, Browne JK, Egrie JC, Nissenson AR, Okamoto DM, et al. The effects of normal as compared with low hematocrit values in patients with cardiac disease who are receiving hemodialysis and epoetin. N Engl J Med. 1998;339:584-90. [PMID: 9718377]

61. Bellinghieri G, Savica V, De Gregorio C, De Gregorio G. Use of erythropoietin in ischemic and arrhythmic cardiopathy of hemodialyzed patients. Contrib Nephrol. 1994;106:135-7. [PMID: 8174359]

62. Palazzuoli A, Silverberg D, Iovine F, Capobianco S, Giannotti G, Calabrò A, et al. Erythropoietin improves anemia exercise tolerance and renal function and reduces B-type natriuretic peptide and hospitalization in patients with heart failure and anemia. Am Heart J. 2006;152:1096.e9-15. [PMID: 17161060]

63. ATS Committee on Proficiency Standards for Clinical Pulmonary Function Laboratories. ATS statement: guidelines for the six-minute walk test. Am J Respir Crit Care Med. 2002;166:111-7. [PMID: 12091180]

64. Anker SD, Comin Colet J, Filippatos G, Willenheimer R, Dickstein K, Drexler H, et al; FAIR-HF Trial Investigators. Ferric carboxymaltose in patients with heart failure and iron deficiency. N Engl J Med. 2009;361:2436-48. [PMID: 19920054]

65. Okonko DO, Grzeslo A, Witkowski T, Mandal AK, Slater RM, Roughton

M, et al. Effect of intravenous iron sucrose on exercise tolerance in anemic and nonanemic patients with symptomatic chronic heart failure and iron deficiency FERRIC-HF: a randomized, controlled, observer-blinded trial. J Am Coll Cardiol. 2008;51:103-12. [PMID: 18191732]

66. Toblli JE, Lombraña A, Duarte P, Di Gennaro F. Intravenous iron reduces NT-pro-brain natriuretic peptide in anemic patients with chronic heart failure and renal insufficiency. J Am Coll Cardiol. 2007;50:1657-65. [PMID: 17950147]

67. Murphy MF, Wallington TB, Kelsey P, Boulton F, Bruce M, Cohen H, et al; British Committee for Standards in Haematology, Blood Transfusion Task Force. Guidelines for the clinical use of red cell transfusions. Br J Haematol. 2001;113:24-31. [PMID: 11328275]

### Ad Libitum

#### He Lectures at the Heritage Association Dinner

After dessert, the women arranged themselves upstairs in a private room. He began by presenting the usual distinction between letting a patient die and killing. His examples generated a spark that ignited tiny firecrackers behind a dozen faces and waving hands rose, Doctor! Doctor! A husband suffering torture long after his body gave out-a physician's hubris. An injured brother severed from machines too soon by doctors who had stolen his wife's consent. Passion took hold of the group. Anger and urgency tore great hunks of experience from the women's hearts and grief uncovered its roots. Having sprung the hinges of ethics, he sampled the best dose in his cabinetsilence. When passion subsided, the president praised the talk that had not been given and thanked him profusely for sharing his wisdom.

*Jack Coulehan, MD, MPH* Setauket, New York

Current Author Address: Jack Coulehan, MD, MPH, e-mail, jcoul44567@aol.com.

© 2013 American College of Physicians

Current Author Addresses: Drs. Qaseem and Starkey: 190 N. Independence Mall West, Philadelphia, PA 19106.

Dr. Humphrey: Oregon Health and Science University, 3710 SW U.S. Veterans Hospital Road, Portland, OR 97201.

Dr. Fitterman: Huntington Hospital, 270 Park Avenue, Huntington, NY 11743.

Dr. Shekelle: West Los Angeles Veterans Affairs Medical Center, 11301 Wilshire Boulevard, Los Angeles, CA 90073. Author Contributions: Conception and design: A. Qaseem, N. Fitterman, P. Shekelle.

Analysis and interpretation of the data: A. Qaseem, L.L. Humphrey, N. Fitterman, M. Starkey.

Drafting of the article: A. Qaseem, N. Fitterman, M. Starkey.

Critical revision of the article for important intellectual content: A. Qaseem, L.L. Humphrey, N. Fitterman, M. Starkey, P. Shekelle.

Final approval of the article: A. Qaseem, L.L. Humphrey, N. Fitterman, P. Shekelle.

Statistical expertise: A. Qaseem.

Administrative, technical, or logistic support: A. Qaseem, M. Starkey. Collection and assembly of data: A. Qaseem.

#### CORRECTION

# Correction: Treatment of Anemia in Patients With Heart Disease

The figure in a recent guideline (1) was incorrect. The correct version appears below.



#### Summary of the American College of Physicians Guideline on Treatment of Anemia in Patients With Heart Disease

Disease/Condition	Anemia and heart disease		
Target Audience	Internists, family physicians, and other clinicians		
Target Patient Population	Adult patients with symptomatic CHF (with or without reduced systolic function) or CHD (acute coronary syndrome, postacute coronary syndrome, or a history of MI or angina) and anemia or iron deficiency		
Interventions	Red blood cell transfusion, ESAs with or without iron (including erythropoietin and darbepoetin), and intravenous iron		
Outcomes	Mortality (all-cause and disease-specific); cardiovascular events (MI, CHF exacerbation, arrhythmia, or cardiac death); exercise tolerance (any metric, most commonly NYHA class, 6-min walk test); quality of life; hospitalization (all-cause and disease-specific); and harms, including hypertension, venous thromboembolic events, and ischemic cerebrovascular events		
Benefits of Treatment	Red blood cell transfusion: no benefits shown when comparing liberal to restrictive transfusion		
	ESAs: no benefit		
	Intravenous iron: increased exercise tolerance, improved quality of life		
Harms of Treatment	Red blood cell transfusion: adverse events potentially associated with red blood cell transfusions, such as fever, transfusion- related acute lung injury, and congestive heart failure		
	ESAs: hypertension and venous thrombosis		
	Intravenous iron: no harms identified but sparsely reported		
Recommendations	Recommendation 1: ACP recommends using a restrictive red blood cell transfusion strategy (trigger hemoglobin threshold of 7–8 g/dL compared with higher hemoglobin levels) in hospitalized patients with coronary heart disease. (Grade: weak recommendation; low-quality evidence)		
	Recommendation 2: ACP recommends against the use of erythropoiesis-stimulating agents in patients with mild to moderate anemia and congestive heart failure or coronary heart disease. (Grade: strong recommendation; moderate-quality evidence)		
High Value Care	Current evidence does not support the benefit of liberal blood transfusions in patients with asymptomatic anemia and heart disease. Therefore, the ACP does not support the liberal use of blood transfusions in the management of mild to moderate anemia in patients with cardiovascular disease. The probability that transfusion may be beneficial is higher in patients with lower hemoglobin levels (<7 g/dL) and lower in less anemic patients (hemoglobin >10 g/dL) (67). The ACP does not support the use of ESAs for treating patients with mild to moderate anemia and heart disease because the harms outweigh the benefits for these patients.		
Clinical Considerations	Patients with heart disease may have anemia because of iron deficiency, use of ACE inhibitors, renal insufficiency, and poor nutrition.		
	Presence of anemia is associated with increased mortality and morbidity. However, it is uncertain if anemia is an independent risk factor for poor outcomes or if it is a marker of more severe illness.		
	The impact of oral administration of iron and how it compares with IV iron for treating anemic patients with heart disease is unknown.		

This has been corrected in the online version.

#### Reference

1. Qaseem A, Humphrey LL, Fitterman N, Starkey M, Shekelle P, for the Clinical Guidelines Committee of the American College of Physicians. Treatment of anemia in patients with heart disease: a clinical practice guideline from the American College of Physicians. Ann Intern Med. 2013;159:770-9.